Author's response to reviews

Title: Takotsubo cardiomyopathy in an adolescent affected by pancreatic adenocarcinoma

Authors:

   Maria Debora De Pasquale (mdebora.depasquale@opbg.net)
   Angela Mastronuzzi (angela.mastronuzzi@opbg.net)
   Luigi De Sio (desio@opbg.net)
   Annalisa Serra (annalisa.serra@opbg.net)
   Chiara Grimaldi (chiara.grimaldi@opbg.net)
   Marcello Chinali (marcello.chinali@opbg.net)
   Ugo Giordano (ugo.giordano@opbg.net)

Version: 5
Date: 25 January 2016

Author's response to reviews:

Dear Reviewer

We would like to thank you for your suggestions.

Changes have been made in the manuscript, according to comments. All changes have been bolded in the text

1 – Title has been changed, according to Your suggestion. In the discussion, sentence about Tako-tsubo like cardiomyopathy has been modified as well.

2- second video has been uploaded in a different video format, in order to be available for vision.